Author’s response to reviews

Title: Factors that influence dietary behavior toward iron and vitamin D consumption based on the theory of Planned Behavior in Iranian adolescent girls

Authors:

Ali Alami (alialami65@hotmail.com)
Seyedeh Belin Tavakoly Sany (belintavakoli332@gmail.com)
Elaheh Lael-Monfared (elm.monfared@gmail.com)
Gordon Ferns (g.ferns@bsms.ac.uk)
Maryam Tatari (tatarim1397@gmail.com)
Zahra Hosseini (Hoseini.z1397@gmail.com)
Alireza Jafari (jafari.ar94@gmail.com)

Version: 2 Date: 16 Jan 2019

Author’s response to reviews:

Dear Editor-in-Chief/ Professor/Dr

We would like to thank you for your positive feedback and for the thoughtful comments and constructive suggestions, which help to improve the quality of this manuscript. This article was edited again for proper English language, grammar, punctuation, spelling, and overall style by Professor Gordon Ferns who is the native authors of this manuscript. All substantive changes made in the revised text highlighted with yellow color.

Best regards

The authors